Author's response to reviews

Title: Congenital syphilis, still a reality in 21st century: a case report

Authors:

Monica Chaudhary (drmonica74@gmail.com)
Bineeta Kashyap (dr_bineetakashyap@yahoo.co.in)
Preena Bhalla (preenadr@gmail.com)

Version: 6 Date: 15 September 2007

Author's response to reviews:

to,
the Editor
JMCR

Subject: Formatting changes

Sir,

I am submitting the revised version of my manuscript with the following changes
1. a space separating the different sections of the title page has been incorporated
2. the authors' qualifications/titles (e.g. PhD / MD / DR / Head of xx) have been removed from the manuscript and submissions system.
3. Telephone and fax numbers for the corresponding author have been deleted.
4. page breaks between sections following the Abstract have been removed.

thanking you
Dr. Monica Chaudhary